CSE 378 Win ter 1998

Mac hine Organization and Assem bly Language Programming

Problem Set #1

Due: W ednes da y , Jan uary 14

1. By the end of the �rst w eek, y ou should ha v e read Chapter 1, Chapter 4 (Sections 4.1 to 4.3)

and Chapter 3 (Sections 3.1 to 3.4)

2. In Chapter 1, do, but do not hand-in, Exercises 1.1 to 1.26

3. In Chapter 4 do the follo wing exercises

� Ex. 4.1 and 4.2. Giv e the results in binary and in hexadecimal

� Ex. 4.4, 4.5 and 4.6. If y ou �nd it con v enien t y ou can use a notation lik e 2 x

+ y .

� Ex. 4.11

4. Assume a 4-bit register and a 2’s complemen t represen tation of in tege rs. Giv e examples of

adding t w o p ositiv e n um b ers with and without o v er o w and of of subtracting a negativ e

n um b er from a p ositiv e n um b er with and without o v er o w. Sho w the 2’s complemen t repre-

sen tatio n of the op erands and the result in a manner similar to what is done in the b o ok pp

220-2 21 .